Author’s response to reviews

Title: Patient- and delivery-system factors related to acceptance of HIV counseling and testing services among tuberculosis patients in South Africa: a qualitative study with community health workers and program managers

Authors:

J. Christo Heunis (heunisj.hum@ufs.ac.za)
Edwin Wouters (edwin.wouters@ua.ac.be)
Wynne E Norton (wenorton@uab.edu)
Michelle C Engelbrecht (engelmc@ufs.ac.za)
N. Gladys Kigozi (kigozign@ufs.ac.za)
Anjali Sharma (anjali.sharma@downstate.edu)
Camille Ragin (camille.ragin@downstate.edu)

Version: 8 Date: 28 February 2011

Author’s response to reviews:

Apologies for the errors. Attached is the revised version of the manuscript, with the following changes made per February 27th, 2011 email from the Editorial Team:

--The title in the manuscript is now identical to the title listed in the submission system
--The list of authors is now identical to the list in the submission system
--The abstract is now identical to the abstract in the submission system
--The references are now in proper consecutive order.